Author's response to reviews

Title: Effectiveness of Isopropyl Myristate/Cyclomethicone Solution of Removing Cuticular Hydrocarbons from Human Head Lice (Pediculus humanus capitis)

Authors:

Eric C Barnett (ebarnett@piedmontpharma.com)
Kathleen G Palma (kpalma@piedmontpharma.com)
Bert Clayton (bclayton@piedmontpharma.com)
Timothy Ballard (tballard@en-cas.com)

Version: 3 Date: 17 August 2012

Author's response to reviews:

Version: 2 Date: 16 August 2012

Comments:
In response to Reviewer report by Ian Burgess; Version 2 4 August 2012, we have made the following changes.
1. Section: Increasing Resistance to Pesticides. We have significantly reduced this section as recommended by the reviewer.
2. Citation errors have been corrected.
3. Section: A challenge to cure. The effect on the nymph has been clarified.
4. GC flame mentions appear appropriate and no further changes have been made.
Minor Revisions. All references to Hedrin have been removed. Spelling errors have been corrected.
The Abstract has been expanded to include Methods